Author's response to reviews

Title: Study Protocol for the Development of a European Measure of Best Practice for People with Long Term Mental Illness in Institutional Care (DEMoBinc).

Authors:

Helen Killaspy (h.killaspy@medsch.ucl.ac.uk)
Michael King (m.king@medsch.ucl.ac.uk)
Christine Wright (cwright@sgul.ac.uk)
Sarah White (swhite@sgul.ac.uk)
Paul McCrone (p.mccrone@iop.kcl.ac.uk)
Thomas Kallert (thomas.kallert@tu-dresden.de)
Jorge Cervilla (jacb@ugr.es)
Jiri Raboch (raboch@nbox.cesnet.cz)
Georgi Onchev (georgeonchev@hotmail.com)
Roberto Mezzina (roberto.mezzina@ass1.sanita.fvg.it)
Durk Wiersma (Durk.Wiersma@med.umcg.nl)
Andzrej Kiejna (akiejna@psych.am.wroc.pl)
Dimitri Ploumpidis (diploump@med.uoa.gr)
Jose Miguel Caldas de Almeida (caldasjm@fem.uml.pt)

Version: 4 Date: 27 May 2009

Author's response to reviews:

Dear Editor

We have made all the amendments to the formatting of the manuscript as requested.

Thank you for accepting our study protocol for publication in BMC Psychiatry.

Kind regards

Helen Killaspy